



Sotiris Angelis^{1,*}, Konstantinos Kotis¹ and Dimitris Spiliotopoulos²

- ¹ Intelligent Systems Lab, Department of Cultural Technology and Communication, University of the Aegean, University Hill, 81100 Mytilene, Greece; kotis@aegean.gr
- ² Department of Management Science and Technology, University of the Peloponnese, 22100 Tripoli, Greece; dspiliot@uop.gr
- * Correspondence: sotiris@aegean.gr; Tel.: +30-697-971-0389

Abstract: Semantic trajectory analytics and personalised recommender systems that enhance user experience are modern research topics that are increasingly getting attention. Semantic trajectories can efficiently model human movement for further analysis and pattern recognition, while personalised recommender systems can adapt to constantly changing user needs and provide meaningful and optimised suggestions. This paper focuses on the investigation of open issues and challenges at the intersection of these two topics, emphasising semantic technologies and machine learning techniques. The goal of this paper is twofold: (a) to critically review related work on semantic trajectories and knowledge-based interactive recommender systems, and (b) to propose a high-level framework, by describing its requirements. The paper presents a system architecture design for the recognition of semantic trajectory patterns and for the inferencing of possible synthesis of visitor trajectories in cultural spaces, such as museums, making suggestions for new trajectories that optimise cultural experiences.

Keywords: semantic trajectories; recommender systems; big data analytics; user experience; cultural space

1. Introduction

Visitors of cultural spaces (e.g., museums, archaeological sites) are usually offered a rather static and less personalised experience, e.g., a group-organised guided tour of exhibits in museum rooms. To overcome this problem, there have been numerous studies that utilise advancements in recent technologies, such as IoT and pervasive computing technologies, to monitor and analyse visitor movement and interactions within cultural spaces [1,2]. Analysed data (with high volume, velocity, and variety) gathered from sensors (streaming/dynamic data) and datastores/databases (historical/static data) are used to recognise/infer visitor preferences and personal interests to propose and eventually deliver an enhanced cultural experience. This is achieved either by providing personalised and enriched content or by suggesting personalised navigation in the cultural space. A related example is found at the Rijksmuseum Amsterdam, which offers a real-time routing system for implementing a mobile museum tour guide for personalised tours [3].

An already effective approach for discovering preferences and needs for moving users in cultural spaces is through the analysis of their trajectories (big movement data), as they contain rich explicit and implicit information and knowledge. Due to the evolution of mobile computing, wireless networking, and related technologies, such as GPS, mobile applications can monitor and share information about user position during movement, e.g., while the user is visiting a cultural space. The existing infrastructure enables applications to produce a vast amount of streaming data that include information not only about locations and places that users are visiting but also the paths/routes/trajectories the users are following, as an aggregation of connected spatial points in specific time-lapses [4–7].



Citation: Angelis, S.; Kotis, K.; Spiliotopoulos, D. Semantic Trajectory Analytics and Recommender Systems in Cultural Spaces. *Big Data Cogn. Comput.* 2021, *5*, 80. https://doi.org/ 10.3390/bdcc5040080

Academic Editor: Giuseppe Maria Luigi Sarnè

Received: 2 October 2021 Accepted: 13 December 2021 Published: 16 December 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). Semantic Web technologies offer powerful representation tools for pervasive applications. The convergence of location-based services and Semantic Web standards allows easier interlinking and semantic annotation of trajectories, resulting in semantic trajectories. Trajectory-based operations, which involve spatiotemporal data of moving entities, are becoming increasingly important in related studies and applications, as they provide insights about human movement and the ability to extract patterns and predict future behaviours. As described in [8], a semantic trajectory-based recommender system (RS) is designed on the basis of the observation that users with similar trajectories would have similar preferences for the available objects, and outperform traditional recommendation methods that do not consider trajectory or environment information.

The motivation for this research is to explore human movement and behaviour in cultural spaces to provide optimised cultural experiences. Based on this motivation, we propose a framework that represents visitor movement as enriched semantic trajectories to extract useful information required as input to a recommender system that would provide optimum alternatives to their cultural experiences. The main requirements of the framework (described in Section 5) are summarised as follows:

- a. Exploitation of raw spatiotemporal trajectory data
- b. Semantic segmentation and annotation of the trajectory
- c. Trajectory description using suitable ontologies
- d. Semantic trajectory enrichment with Linked Open Data (LOD)
- e. Semantic annotation of cultural spaces and points of interest (POI) to provide context and capability for semantic integration with user trajectories
- f. Trajectory analytics for pattern recognition and classification
- g. Future location prediction
- h. Dynamic user profiling
- i. Integration of User Knowledge Graph (UKG)
- j. Integration of Cultural Space (CS) and POI Knowledge Graph (KG)
- k. Integration of KG-Based recommender system (RS) for path-based and KG-based recommendations
- l. Integration of context-aware RS
- m. Integration of hybrid RS
- n. Integration of collaborative filtering RS
- o. Inference and proposal of a possible synthesis of visitor trajectories

The specific set of requirements was derived from an extensive study of the related work (pros and cons of related approaches/systems), as well as from our motivation to select the most appropriate techniques and methods related to the synthesis of both paradigms, i.e., semantic trajectories and recommender systems. The aim was to develop a novel framework that will eventually (a) enhance the functionality and efficiency of a recommender system for visiting cultural spaces, and (b) connect user trajectories with optimised cultural experiences.

The aim of this paper is two-fold: (a) to present a systematic literature review of state-of-the-art approaches related to semantic trajectories and recommender systems, with an emphasis on the cultural domain, and (b) to introduce a high-level framework for the recognition of trajectory patterns, inferencing possible syntheses of visitor trajectories in cultural spaces and the combination of trajectory analysis results with visitor dynamic profiling data. The RS is used to suggest optimal alternative trajectories in real-time to enhance the visitor experience. The framework is presented through a use case scenario of a trajectory that takes place in the city of Athens and the Acropolis Museum.

The structure of this paper is as follows: Section 2 describes the basic concepts of Semantic Trajectories, Recommender Systems, and Knowledge Graphs. Section 3 discusses the survey methodology and the decisions made for the final selection of related studies. Section 4 presents the reviewed state-of-the-art related work regarding Semantic Trajectories and Recommender Systems. Section 5 critically discusses the related work based on a set of requirements towards a framework that enhances cultural experiences. Section 6

presents the proposed system architecture design. Finally, Section 7 concludes the paper and reports on future work.

2. Preliminaries

2.1. Semantic Trajectories (STs)

A trajectory is defined as the composition of the sections of connected traces and points that express a meaningful movement in space and time by an object or entity of interest. The study of trajectories is fundamental for the comprehension of moving object/entity behaviour, as there is a plethora of useful information in the path/route the object/entity follows to navigate between start and end points. The behaviour of a trajectory is the sum of the characteristics that identify the essential details of a moving object/entity or a group of moving objects/entities. A set of such unique characteristics creates a short description of a group of trajectories which are called patterns [9,10]. Data analytics based on trajectories of moving objects/entities, such as trajectory clustering and construction, could provide advantages in the solutions of several common or more complex problems [11–13]. Trajectory analysis can be performed either using raw spatiotemporal data or semantically annotated movement data. Although data mining plays a significant role in this domain, as these algorithms are used for the extraction of trajectory patterns, most of the mining algorithms are developed for raw data implementation and, as a result, they are not effective in recognising patterns for specific domains. Moreover, the exclusive usage of spatiotemporal data provided by movement tracking sources lacks significant information about the context of the movement [4,14].

One of the main issues is the difficulty to correlate the patterns with movement behaviours to extract and expand the knowledge about them. The challenge is in the enrichment of the spatiotemporal data of the trajectories with semantic, context-based information, relevant to the moving objects and the contextual association of the trajectories with related low- or high-level events. One way to address this challenge is the use of ontologies and linked data (LD) to semantically connect the trajectory patterns and behaviours with the broader context of the movement. As stated by Dodge et al. [15], the movement behaviour depends on the general context in which it takes place, as every movement has a specific meaning in the moment and in the space/environment that it is happening. Semantic trajectories are the trajectories that have been enriched with semantic annotations and one or more complementary segmentations [10]. Annotations of segmented parts of a trajectory (episodes) could be "stop" or "move", or, in other cases, could be points or regions of interest. An example semantic trajectory of a touristic walk is depicted in Figure 1.

Knowledge discovery tasks can be performed in semantic trajectories to extract patterns based on characteristics, such as changes or stops, POIs, or specific behaviours that can be recognised in a single or a group of trajectories and used to create classes and classify or identify future trajectories [9].

2.2. Recommender Systems (RS)

RSs are software tools or AI applications that are designed to predict the user's interests and preferences based on statistics, data mining algorithms, and machine learning techniques, to suggest/recommend products, services, locations, routes, etc. RSs handle the problem of information overload that users normally encounter and affect the way users make decisions through the recommendation of suitable actions or objects of interest. An example of a Cultural Recommender System is depicted in Figure 2. The main tasks of a RS are to (a) gather and process data, (b) create models based on the available data, (c) apply the models to existing and future data, and (d) receive feedback and re-evaluate the models. Common characteristics that a RS should possess for it to be considered efficient are: accuracy, coverage, relevance, novelty, serendipity, and recommendation diversity [16]. Most recommendation strategies are defined based on three main relation types: (a) User–Item relation, (b) Item–Item relation and (c) User–User relation.



Figure 1. Example semantic trajectory depicting a tourist behaviour in Athens, starting from a simple trajectory, and resulting in a semantic trajectory.



Figure 2. Example of a Cultural Recommender System.

- User–Item relation: This relation is based on the user profile and the explicitly documented preferences of the user towards a specific type of Item.
- Item–Item relation: This relation occurs based on the similarity or the complementarity
 of the attributes or descriptions of the Items.
- User–User relation: This relation describes the Users that possibly have similar tastes with respect to specific Items, such as mutual friends, age group, location, etc.

User preferences are calculated with the use of explicit ratings provided by user evaluations or feedback, and implicit ratings that are inferred from the users' interactions with the products or the aforementioned relations.

Depending on the available data, there are two main methods of categorisation of objects (supervised and unsupervised) based on the similarity of their features. Classification is the supervised method. It uses predefined tags and classes to categorise a group of inputs. Commonly used algorithms for classification are K-Nearest Neighbours (kNN), Decision Trees, and Naïve Bayes. In the case of Clustering, the unsupervised method, labels or categories are unknown in advance and the task is to effectively categorise given inputs and discover similarities based on certain criteria. Examples of clustering algorithms are K-Means Clustering and DBSCAN (Density-based Spatial Clustering).

One of the most common issues that RSs deal with is the rating sparsity, known also as the cold start problem, which causes inefficiency and low accuracy on recommendations. RSs face that condition when the number of available items, i.e., candidates for recommendation, is much greater than the number of users rating them. This situation occurs especially when a RS has recently started collecting ratings or has not been exploited by a lot of users [17].

Popularity-based RSs are often used to bypass the cold start problem that recommend the most selected products by other users. Demographic RSs, use information like age, gender, etc., to classify users for future recommendations. Demographic techniques are sometimes included in Hybrid RSs to increase robustness [18].

Collaborative Filtering (CF) RS use models that focus on User–User relations or Item– Item relations to infer ratings about products. The methods used in CF are equivalent to those of a classifier that creates a training model from labelled data. The basic idea of CF methods is that unspecified ratings can be imputed because the observed ratings are often highly correlated across various users and items. The main challenge in those methods is that the matrices of the user ratings are very sparse, as the users usually rate a fraction of the available products. When the user preferences are specified, the model tries to discover similarities to other users. If the similarity discovery is successful, the ratings of similar users are used to infer values to complete the rating matrices [16].

Content-based RSs are based on the idea that the user will prefer products similar to those already used and rated highly. This type of RS creates representations of products based on their features and descriptions and matches them to similar attributes of other products to suggest them to the target user. The content-based methodology calculates user profiles and specifies the interests and preferences to compute a relevance score that predicts the user's level of interest in a specific product. The product attributes for representations are usually extracted by metadata or textual descriptions, but there is an increasing interest in the advantages of Semantic Web technologies to approach content-based recommendations. As there is a plethora of open knowledge sources that provide semantic information, recent research studies shift from keyword-based to concept-based representations of products and users [19].

Apart from the Item ratings, context may be anything that might affect the desirability of particular recommendations at the time of the generation of the recommendations [19]. The context-aware RS (CARS) is a new trend in recommender systems. The CARS considers user profiles as dynamic and evaluates user preferences and interests along with other factors that may occur in the current situation of the target user, like the user's location, the user's companionship, the weather, etc. The CARS aims to provide personalised recommendations according to both user profiles and their current contextual conditions [20].

The knowledge-based RS utilises domain knowledge either provided by experts in the form of domain-specific rules and ontologies or by the usage of the knowledge available on the Web as structured LOD. KGs can be used to exploit explicit connections between user entities and product entities or infer implicit connections to create suggestions for the users. As mentioned in [21], there are several studies on ontology-based, LOD-based, path-based, and KG-based recommendation approaches that perform better than traditional recommendation approaches, especially in cases with small amounts of sample ratings and sparse rating matrices.

The hybrid RS exploits the advantages of different approaches, like the effectiveness of KG-based RSs in sparse data, or the collaborative methods when multiple user ratings are available. The hybrid RS leverages the strengths of several approaches, allowing recommendation methods to produce separate ranked lists of recommendations, and then merging their results to produce a suggestion list.

2.3. Knowledge Graphs

KGs are increasingly getting attention from academic and industry organisations as they provide several advantages compared to relational databases, regarding the representation and management of big and heterogeneous data. As defined by Hogan [22], a KG is a graph of data intended to accumulate and convey knowledge of the real world. The nodes represent entities of interest, and the edges represent relations between these entities. While there is a conceptual overlap between KGs and ontologies, because both are formed to be "an explicit specification of a conceptualisation", KGs can be considered more as "a graph of data with the intent to compose knowledge" [23,24].

As described in [24], the requirements for a graph to be considered a KG are:

- a. Knowledge graph meaning is expressed as structure.
- b. Knowledge graph statements are unambiguous.
- c. Knowledge graphs use a limited set of relation types.

In terms of inferencing and entity representation, a KG can accumulate simple statements as edges in the data graph, but for more advanced tasks, more expressive ways are required, such as the use of ontologies and rules [22].

A KG stores and manages relations about entities. These relations could be expanded, and new relations could be created with the use of inference algorithms, e.g., rule-based reasoning, OWL/RDFS reasoning, or combinations of these approaches, that can infer knowledge and enrich the KG.

Figure 3 depicts an example KG that includes a user sub-graph describing the visitors, a POI sub-graph describing museums and exhibits in the city of Athens, and their connections. A sample set of RDF triples [25] of this KG is the following:

:Peter	rdf:type	foaf:Person.
:Peter	foaf:knows	:Mary.
:Mary	:visited	db:Parthenon.
db:Parthenon	rdf:type	:POI.
db:Parthenon	dbo:location	dbr:Athens.

Apart from KG enrichment, several techniques can be applied to a KG to provide insights and perform analysis of the encoded data. The most widely used techniques and approaches for analysis are listed below:

- Centrality: discovers the nodes with the most connections and the biggest impact in the graph.
- Community Detection: discovers sub-graphs that are more closely connected internally, compared to the rest of the graph.
- Connectivity: evaluates the quality of the connections in the graph, in terms of resilience, reachability, etc.
- Node Similarity: measures which neighbour nodes are in a specific area of the graph, based on their features and connections.
- Path Finding: discovers possible reachable paths between predefined terminal nodes.

- KG Embeddings: transforms graph representations to a low-dimensional vector space (graph embeddings), to allow ML applications to handle them efficiently.
- KG Recommendations: KGs, by design, provide the technical means to integrate various heterogeneous information sources, for instance, POIs and user preferences. Thus, feature similarity discovery algorithms can be applied to enhance recommendation techniques.





3. Survey Methodology

The research methodology followed in this paper focuses mainly on the collection of information sources related to semantic trajectories and cultural recommender systems. The research was conducted in a period of six months, examining academic articles, relevant literature, and web resources published between 2016 and 2021 (5 years).

The research for articles was conducted on academic web portals such as Scopus, Google, and Semantic Scholar, ResearchGate, ACM Digital Library, IEEEXplore, and SpringerLink.

The specific search terms used in various combinations were:

- semantic recommender systems
- trajectory-based recommender systems
- semantic trajectory-based recommender systems
- cultural recommender systems
- semantic trajectories
- trajectory annotation
- trajectory segmentation
- POI extraction and annotation
- trajectory enrichment
- human movement trajectories
- cultural semantic trajectories

The papers that were collected were mainly related to the domains of museum studies, cultural computing, and computing.

We have included articles that mostly cover human movement and trajectories that are primarily related to the domains of our field of study. Although a significant amount of the published research is on semantic trajectories and trajectory analytics in a wide range of scientific fields, we did not include works applied to other domains, such as the works related to human road safety or autonomous vehicles. However, we have taken into consideration a selection of non-human trajectory-related studies that propose noteworthy approaches regarding the semantic annotation and management of trajectories [13,26–37].

RSs are increasingly applied to a variety of use cases. Therefore, there are numerous articles about their use in several domains. We have limited the selected works to those that utilise knowledge and semantic approaches for recommendations and those that focus on cultural applications and trajectories [2,8,38–51].

4. State of the Art in STs and RSs

4.1. Semantic Trajectories (ST)

4.1.1. Annotation of Trajectories

In de Graaff et al. [26], a novel algorithm for automating the detection of visited POIs is proposed. They describe a POI as "a location where goods and services are provided, geometrically described using a point, and semantically enriched with at least an interest category". A Polygon of Interest (POLOI) has a similar definition to a POI, except that the location is described by a polygon. The proposed method tries to identify visited POIs both in outdoor and indoor trajectories from raw smartphone GPS data but is specifically designed for urban indoor trajectory analysis. It focuses on detecting stops that take place at known and predefined POIs. The challenges overcome by the proposed algorithm are the non-detection of indoor visited POIs and the false positive detection due to lack or instability of the GPS signal. Because of the unavailability of the GPS signal for the indoor segments of a trajectory, the proposed algorithm selects points before and after the users get into a building and projects them to a polygon. The POI visit extraction algorithm considers the accuracy of the location, reductions in speed, changes in direction, and projection of signals onto polygons to extract the staypoints (the centroids of stop sequences) from a trajectory. An experiment with students in the city of Hengelo was set up to validate the proposed approach and concluded that the algorithm outperformed several existing approaches.

Another method for annotating trajectories, following a different approach, is the one presented by Nogueira et al. [13]. In this work, it is stated that the focus is on modelling mobile object trajectories in the context of the Semantic Web, and it is based on an ontological approach to extract episodes from semantic trajectories. An episode is defined as the smallest semantic unity in Semantic Trajectory Episodes (STEP) [52]. Episodes encapsulate values that a Feature of Interest (FOI) or Contextual Element (CE) may assume during the trajectory. They expand the already published STEP ontology to represent generic spatiotemporal episodes. It is claimed that, with the use of Semantic Web technologies, it is possible to integrate various data sources, data, and metadata and also support reasoning by inference, in order to enrich movement data. To validate the expanded ontology, a framework called FrameStep is proposed. The framework contains a graph–object mapping layer that enables the creation of episode instances that can be then transformed and serialised into triples. It utilises the ontology along with annotation algorithms to form semantic trajectories from raw GPS data by detecting episodes and enriching the trajectories. The framework also integrates with LD cloud and OpenStreetMap [53] to retrieve information about the context and the environment of the analysed trajectories.

In the work of Chen et al. [30], the challenge of organising raw spatial trajectories and fusing them with semantic data is examined. They present a structured and selfdescribed way to annotate trajectory episodes with sentiments, events, or topic words. To achieve this goal, a model is implemented, where each user has multiple trajectories divided into episodes. Each episode contains semantic information like events or sentiments extracted by a combination of spatial and temporal information, with the context semantic information derived from posted texts by the users, using Natural Language Processing (NLP).

4.1.2. Semantic Trajectory Management

The work of AL-Dohuki et al. [31] focuses on an approach to interact with trajectory data through visualisations, enriched with semantic information about the trajectories. The approach was designed and evaluated for taxi trajectories. The trajectories are converted to documents through a textualisation transformation process where the GPS points are mapped to street names or POIs, and the speed is described quantitatively. After the transformation, each document is described by a meta-summary and indexed to enable queries over a text-based search engine. The system is data-structure agnostic, and the results are integrated with visualisations and interactions to promote easy understanding. The evaluation of a prototype claimed to be successful, and its ease of use is demonstrated appropriately.

Motivated by the need to exploit data from disparate and heterogeneous sources in an integrated manner to increase the predictability of moving object trajectories, Santipantakis et al. [32] propose a framework for semantic integration of big mobility data with other data sources, providing a unified representation and support analysis tasks by exploiting trajectories at various levels of analysis. One of the major challenges of this work is to enrich surveillance data providing meaningful information about moving entity trajectories, and annotating trajectories with related events, therefore creating enriched trajectories. To meet these challenges, they introduce the SPARTAN approach for providing enriched streams of mobility data, incorporating online compression, data transformation, and link discovery functionality. The domains of interest are marine and aviation. Streaming spatiotemporal data are the input to the proposed framework. It performs data cleaning and summarisation, transforms data to Resource Description Framework (RDF) [25] in compliance with a generic ontology (namely, datAcron ontology [54]) for trajectories, and performs integration with other streaming and archival data sources. The output of the framework is a stream of linked RDF data that contains enriched trajectories of moving objects. The experimental evaluation of the proposed framework demonstrates efficiency and scalability when using large, real-life datasets.

The first step to semantic annotation and enrichment of the trajectories is to perform the necessary trajectory segmentation. This can be done either with supervised or unsupervised ML techniques. In the first case, the criteria are predefined, and the segmentation can be implemented with ML techniques trained with labelled segments or by applying algorithms that segment the trajectory based on a threshold. In the latter case, the applied methods must discover similarities and homogeneity without specified criteria, based on point density or a cost function. In the work of Amilcar Soares Júnior et al. [33], a semi-supervised algorithm is proposed that implements the Minimum Description Length (MDL) principle to measure homogeneity inside segments. This algorithm uses a small set of labelled trajectory data during the trajectory segmentation task to drive the unsupervised segmentation of unlabelled trajectory data. The semi-supervised approach takes advantage of both the supervised and unsupervised techniques, so the user must annotate a small set of trajectories to help the algorithm recognise meaningful segments for the rest of the trajectories. As mentioned in the study, the main advantage of this method, compared to pure supervised ones, is that it reduces the human effort to label the number of trajectories. A few examples can be used to target the segmentation task to specific domains. The proposed algorithm is characterised as reactive, for automatically defining the values of the input parameters by analysing the results of previously provided examples. As stated in the study, for the experiments conducted using real-world datasets, the proposed algorithm outperformed the state-of-art competitors.

4.1.3. Semantic Trajectory Modelling

Semantic modelling is the step after the annotation for utilising semantic trajectories. In Vassilakis et al. [34], authors present SemMR, a semantic framework for modelling interactions between human and non-human entities, managing reusable and optimised cultural experiences towards a shared cultural experience ecosystem that may seamlessly accommodate mixed reality experiences. The proposed framework is based on the concept of cultural experience as a semantic trajectory (eX-trajectory). It is designed to utilise Mixed Reality (MR) technologies and applications for creating and managing eX-trajectory content and tools. Methods are proposed for monitoring and analysing user interactions in MR spaces for behaviour pattern extraction for optimising eX-trajectories at runtime by enriching and augmenting them with useful information from heterogeneous sources. To evaluate the framework, simulation experiments ran with artificial users as inputs. As stated, the eX-trajectories that were generated by the system were parsed and evaluated for appropriateness to each user profile and visitor path and were found to be in alignment with the user visiting style and preferences.

The related work for modelling human movement behavioural knowledge from GPS traces for categorising mobile users [35] is motivated by the challenge of using extracted knowledge from trajectory analysis of a specific place to cluster users and discover user behaviour patterns in a target place, without analysing it. That work proposes a framework that forms trajectories from raw GPS data, creates a movement behaviour model, and applies pattern mining methods to transfer knowledge regarding the human movement to geographical regions. To achieve the knowledge transfer, they cluster user behaviour in a specific region of interest and apply the results to semantically similar regions. To model the user movement, a Bayesian network is used that can encapsulate measures of the randomness of movement. The classifier of the Bayesian network provides the probability of a target user to match multiple categories that are predefined based on the features and the trajectory analysis. As an experimental evaluation, a real-life dataset of monitored GPS trajectories was used to transfer knowledge to a region with insufficient data and the results were close to those of the analysed area.

As mentioned by most of the reviewed related works, the mainstream trajectory representation methods focus mainly on the spatiotemporal information of the trajectory and not on the context or the semantic information that could be extracted from them. Gao et al. [36] present a novel representation of semantic trajectories that takes into account domain knowledge, in addition to spatial and temporal data, in order to enhance semantic trajectory retrieval. They propose a synchronisation-based clustering model to transform raw GPS points to multi-resolution Regions of Interest (ROIs). They deploy a tree-shaped hierarchical network that captures each ROI in a set of GPS trajectories. This leads to the replacement of raw trajectories with sequences of ROIs. A hierarchical embedding model transforms the ROI sequences to continuous vectors, based on geographical and semantic trajectory features in a way that the similarity measures between two trajectories can be computed by the Euclidean distance of two vectors directly. The embedding model emphasises their context of movement to extract semantic relations among target objects. The results presented after evaluation experiments showed that the proposed method had superior performance in semantic ROI/trajectory retrieval tasks compared to state-of-theart methods and deep network embedding models.

Most of the studies in the field of trajectory representation and analytics focus on GPS data of outdoor activities and movements, while those that capture indoor POIs present them as parts or stop points of a broader trajectory. Motivated by the lack of indoor trajectory research, Kontarinis et al. [37] combine aspects of semantic outdoor trajectory models with a semantically-enabled hierarchical symbolic representation of the indoor space, in alignment with the OGC IndoorGML standard [55]. The proposed model for enriched indoor semantic trajectories utilises a standardised indoor space modelling framework that contains the semantic trajectories alongside the semantically enriched representations of indoor space. The indoor space is described as a layered multigraph.

The nodes of the multigraphs represent spatial regions, while the edges represent the topological relations between spatial regions. The model was deployed and evaluated for a dataset of spatially aggregated timestamped points of visitors of the Louvre Museum, to present its expressiveness. The study also presents the application of standard and advanced pattern mining methods to provide a formalisation for indoor trajectory mining and express the combination of semantic and spatiotemporal data.

In the work of Krisnadhi et al. [56], a pattern for modelling semantic trajectories is presented. This model adds a spatiotemporal expansion to a previously published model for the representation of trajectories. The work is motivated by the lack of published patterns for simple spatiotemporal extents. The pattern indicates that a trajectory should be modelled as a sequence of fixes that are connected by segments and that every fix should have a spatial and a temporal extent. The model follows a set of axiomatic rules. Specifically, every trajectory must have one starting and one ending fix, trajectories for the same spatiotemporal extent cannot have temporal overlap, and every spatiotemporal extent must have at least one trajectory.

4.1.4. Semantic Trajectory Analytics

While there are several works in trajectory analysis and enriching trajectories and managing them as semantic trajectories provides better insights into the target movement behaviour, limited research has been conducted on the use of Convolutional Neural Networks (CNNs) in connection with modelling human movement patterns. In Karatzoglou et al. [27], a novel CNN-based approach for representing semantic trajectories and predicting future locations is introduced. The CNN approach design was based on an NLP use case similar to the proposed, explaining that the data were also in one-dimensional format and that relevant use cases with the use of CNNs have already been studied in NLP. The CNN takes semantic trajectories as input and assigns a unique index to every semantic location. Trajectory indexes are passed to a hash table which assigns a feature vector to them. These feature vectors are task-specific representations that the system extracted in the training phase from the available data by discovering the optimal semantic location representations. The results are used as input to the core model for predicting the next semantic location. To evaluate the approach, they have worked with semantically enriched data of a single-user model and multiuser models that contained the trajectories of 100 users. The evaluation results showed that, although the proposed model is sensitive to sparse data, it outperforms other reference systems in terms of accuracy and is capable of modelling semantic trajectories and predicting future semantic locations.

In Zhang et al. [28], a system is proposed to extract the semantic trajectory patterns of data produced by users' positioning devices. As stated in the paper, the already proposed mined trajectory patterns are unable to reflect the semantic information hidden in the trajectory because a user's trajectory not only contains the physical movement track but also embodies the user's purpose for moving. Motivated by that, they propose a probabilistic generative model that annotates and clusters the pre-processed trajectory. Raw spatiotemporal data with no semantic information are divided into two parts: the moving and the stop data. They formulated a spatiotemporal threshold and clustering-based method to extract the stopover points. A probabilistic generative model was implemented to identify starting and ending points on a trajectory, connect them to POIs and, by annotating those points, discover the visiting purpose of the trajectory. Finally, the PrefixSpan algorithm [57] is applied to discover patterns over trajectories by finding suffixes for multiple prefix sequences. Evaluating experiments of the proposed method, in contrast to widely used algorithms in pattern mining like K-Means and DBSCAN, showed that it outperforms the other methods in all cases.

Several privacy issues can emerge when tracking and analysing human movement, especially when enriching the movement with context information and discovering patterns that are frequently followed. In order to reduce the risks of invasion of privacy, the work of Khoroshevsky and Lerner [29] is based on the assumption that there is no data sharing

among users and that the user data are stored on the client. By performing the analysis on the user device, they prevent the exchange of sensitive location information between servers and users. An algorithm that combines spatial and semantic information for movement pattern discovery and location prediction is proposed. They retrieve semantic data from the OpenStreetMap API to specify semantic places around points that are clustered in geographic locations. To achieve point clustering in semantic locations, they propose two clustering evaluation metrics. After point clustering, the terminal and intermediate stop points form the user location history, which can be transformed into a sequence of visit locations. To cluster trajectories, they use a similarity metric for string sequences. To discover trajectory patterns, each sequence is assigned to the closest detected cluster of trajectories. Experiments that compared the proposed algorithm with previous works revealed that it provides accuracy for a reasonable number of location sequences.

The work of Liu and Wang [8] is motivated by the challenge of community detection over a set of trajectories. In contrast to most proposed methods that evaluate clustering by proximity-related metrics, they propose a framework that exploits semantic information from multiple sources, where the trajectories in a specific cluster exhibit similarity in one or more movement-related features. They defined the difference between clustering and community detection as the difference between a set of objects related purely through spatial proximity and a set of objects whose proximity or movement similarity is likely a manifestation of some underlying mutual interaction or shared relationship. The proposed framework leverages information markers underlying the raw trajectory data to detect groups based on movement information of users and semantic information of the space where the movement takes place. The framework learns the consistent graph Laplacians by constructing the multi-modal diffusion process and optimising the heat kernel coupling on each pair of similarity matrices from multiple information sources. It models the trajectory similarity based on semantic-level movement, spatiotemporal proximity, and velocity, then computes similarity measurements to determine the communities derived from the computed values. After communities are formed, they apply a collaborative filtering recommendation method based on the observation that similar users that belong to the same community have similar preferences. After experiments were conducted on selected datasets, it is stated that the community detection system and the recommendation method outperformed other clustering and recommendation algorithms.

4.2. Recommender Systems (RS)

4.2.1. Cultural RS

In Amato et al. [44], a methodology that combines recommendation with agent-based planning techniques to implement a planner of routes within cultural spaces is proposed. The problem of finding a scheduled path of visitors in a cultural space or site is handled as a reachability problem and uses multi-agent models to achieve the goal of accessing POIs within certain deadlines. First, the approach analyses user preferences to provide an accurate list of cultural items. Then, the multi-agent planning methods calculate the paths that follow sequence steps to meet the goal of visiting the suggested items. For each pair of users and items, the recommender can compute a rank that measures the expected interest of the user in an item, using a knowledge base and a ranking algorithm. The ranking algorithm integrates information about preferences and past behaviours of the target user and the user community, user feedback, and contextual information, to create the list of suggested items. The browsing system is represented as a directed labelled graph and depicts the sequence of chosen items to increase the similarity measure between them. Finally, the agents compute and recommend the path that meets the requested goals or state that it is unreachable.

Su et al. [45] propose and develop a Big Data architecture that leverages edge intelligence in addition to cloud computing for a more scalable and user-centric analysis of the cultural data. The architecture consists of a data ingestion stage, a knowledge base, a data process stage, and applications. Apart from the architecture, a user-centred recommendation edge-intelligence strategy is proposed for cultural recommendations. The RS relies on a context-aware hybrid recommendation strategy deployed on a multilayer architecture based on Big Data and edge computing technologies. It considers user preferences, location, and items' semantic features of POIs, to generate touristic routes as a sequence of POIs. The recommendation algorithm relies on features of cultural items, the user's past itineraries, and behavioural information captured by the stream processing from social networks. As a proof of concept, they developed an application for suggesting POIs, such as museums, to city visitors. Evaluation results demonstrated that the recommender system outperformed other proposals. This is explained using the Knowledge Base that stores information about objects and users, combining them with those obtained from social networks during the ranking phase.

The work of Cardoso et al. [46] presents the implementation of an application that suggests routes for cultural heritage visits. The application is designed with an adaptive user interface where routing and augmented reality are connected to acknowledge the needs of user categories, such as elders, kids, experts, or general users. The proposed application aims to suggest the optimal route of POIs between terminal nodes in a cultural environment. The suggested route is computed considering the maximum visit time and a vector of the user preferences. The design of the proposed application is based on the optimisation problems of user preference extraction, the number of visiting POIs, and time spent exploring them. The methods designed for the navigation problem were based on the ant colony optimisation algorithms and weighted function strategy. They computed the optimal paths inside a network of POIs, in near real-time, considering the user preferences and given limitations. The application is in the final stages of development and under testing with real users in a real museum environment.

The system presented in Smirnov et al. [47] deals with tasks related to the info-mobility concept: user action analysis, preference revealing, and cultural heritage recommendation based on the preferences and current situation. The system suggests possible touristic paths for visiting cultural POIs. The application architecture is based on the Smart-M3 information sharing platform and consists of a set of joined services by a smart space that provides semantic-based information exchange. The RS calculates the item ranking based on the CF method to provide the list of POIs. The ranking algorithm considers the ratings set by all the users of the system concerning the similarity to the current user, the user preferences, the current situation in the target location, and the reachability of the POIs. The user situation while using the application about cultural heritage POIs that are retrieved from internet sources in real-time. It also estimates the reaching path and presents it in an interactive map. The evaluation shows that the application is efficient in finding the cultural POIs in the user's area in a reasonable time and can be used for on-the-fly tourist support during a trip.

An interesting research challenge is how the cultural factors influence RS efficiency, whether cultural differences could be a technology barrier, or whether there are universal factors for RSs. Motivated by that challenge, Hong et al. [48] proposed the novel concept of cross-cultural contextualisation and a model to compute the cross-cultural factor affecting user preferences. They propose a contextualisation model for computing the cross-cultural factor, which influences user preferences in RSs by using Matrix Factorisation and clustering techniques. As mentioned in the study, a systematic analysis of the dataset and the experimental results suggested that individual users could be considered as country-wise groups for the model to analyse the cross-cultural factors. The users' cultural preferences are modelled to a rating matrix that contains vectors of cultural preferences on different items. Matrix factorisation is applied to interpolate the sparse matrix. Experiments using a real-world dataset, which contains ratings by visitors from different countries, showed the effectiveness of the proposed model and supported that there are cultural factors that influence user rating behaviour in recommendations.

In Loboda et al. [49], the authors demonstrate the impact of the museum RS and implement a content-based RS to generate personalised museum tours to enhance visitors' experience by providing a personalised way to engage with museum collections. The feedback received from a study conducted to evaluate the improvement of a visit by a RS focused on the provided information about the objects and the accessibility of the museum collection. The developed RS was based on the content-based filtering method regarding the feature similarities across the items of the collection. To address the cold-start problem, the users had to select preferred items from a list before the application provides a personalised tour. The application provides detailed information about the recommended items. Evaluation by real visitors reported that the RS made the visit more structured and helped the discovery of interesting objects. Another aspect of the findings of the user evaluation was that diversity in museum RSs might be favoured over accuracy.

In related works about social recommendation services for cultural heritage [50], it is stated that we can guess the affinity of an artwork choice between two users from the users' artwork-watching histories and the artwork features. Motivated by that, Hong et al. presented a novel recommender system based on a method for discovering and exploiting social affinity between users based on artwork features and user experience. The system is designed based on a use case scenario where the museum that exploits the RS is equipped with an IoT system relying on intelligent sensors and services that can identify the user behaviour during the visit. The RS follows a hybrid recommendation approach, where the suggestions are computed from the results of a social-based and context-based recommendation method. The social recommendation ranks the list of artworks based on social affinity by requests of individuals or groups. The affinity between users is estimated using an affinity graph, which is created from filtering the information of the user history and the features of the artworks. For artwork similarity measurement, the Jaccard similarity coefficient and Euclidean distance were calculated. The context-based method recommends artworks based on user interaction. The proposed approach is about to be evaluated through a developed application in a museum in Naples that offered the necessary infrastructure.

4.2.2. Semantic and Knowledge-Based Recommender Systems

Interactive RSs are modelled as a multistep decision-making process to capture the dynamic changes of user preferences. Zhou et al. [38] present a recommendation approach that utilises reinforcement learning methods and KG to provide semantic information to an Interactive RS. Reinforcement learning methods face an efficiency issue when provided with a small sample of data. To address the issue, they leverage prior knowledge of the item relations in the KG for better candidate item retrieval, enrich the representation of items and user states, and propagate user preferences among the correlated items. Interactions between the user and the system last for a defined time period. At each period, the system dynamically generates a list of items based on historical interaction data and item similarity from the KG, suggests them to the user, and receives feedback in order to update the recommendations. The introduced model consists of a graph convolution module, a state representation module, a candidate selection module, and the Q-learning network module. Evaluation experiments demonstrated that the proposed approach outperformed the state-of-the-art method.

In Sansonetti et al. [2], the authors introduce a hybrid RS empowered by social media interactions and LOD. The first step in the recommendation method is the collection of data relevant to the user by analysing the social profiles to retrieve preferences and past interactions with other users and with cultural places. The collected information is stored in a Neo4j [58] graph database and disambiguation tasks are applied to the graph through LOD tools. The proposed recommendation approach integrates both social and semantic recommender methods. The social recommendation method performs CF to suggest items that users with similar interests from the social network prefer. Semantic recommender leverages the DBpedia [59] and Europeana [60] knowledge bases for suggesting cultural

places that share similar semantic features. The system also considers the contextual information of the suggested places, such as accessibility and weather conditions. The recommended POIs are formed in an itinerary with user preferences and contextual constraints. The routes are modelled as a directed graph where nodes represent the POIs, and the weighted edges represent the time required to reach the next POI in the sequence. The graph is filtered based on the location of the POIs. POIs are annotated with information from the LinkedGeoData [61] dataset using SPARQL [62] queries. Experimental results on real users showed the effectiveness of the modules of the proposed RS.

Minkov et al. [39] propose a graph-based recommender framework, to help museum visitors deal with information overload. A KG is constructed to model the museum environment into classes of entities. The nodes represent entities of users, multimedia presentations, physical positions of artworks, or semantic themes, while edges represent structured relations between entities or viewed relations between users and artworks. To infer similarity between nodes, the Personalised Page Rank (PPR) algorithm [63] with a random walk is applied to the graph to rank items based on their relevance to the target user profile. The user profile is generated by using the previous interactions of the user with other entities in the graph. So, user feedback is given on viewed multimedia presentations or POIs while not-yet-watched presentations are ranked up according to the user interactions and preferences. As reported in the study, the results of experiments conducted using data collected at the Hecht Museum showed that graph-based recommendation using the PPR measure outperformed a set of classical collaborative and content-based recommendation methods, justifying the superiority of the graph-based approach.

Qassimi and Abdelwahed [51] present a graph-based recommendation approach that utilises semantic information extracted from collaborative tagging of cultural heritage places to enhance cultural heritage visits and suggest semantically related places that are most likely to be of interest to the visitors. The recommender system is based on the emerging graphs representing the semantic relation of similar cultural heritage places and their related tags. The emerging graphs form a multilayer graph. Its nodes represent the cultural heritage places. The edges represent their relations. Descriptive metadata are extracted by exploring a folksonomy of shared resources to augment the cultural places. The augmented places are clustered based on the tags used to annotate them. The user is provided with suggestions of similar places to those previously visited, tagged, or rated. Evaluation of the system shows that it achieves better results compared to content-based approaches.

4.2.3. Trajectory-Based Recommender Systems for Cultural Spaces

Rodriguez-Hern et al. [40] introduce a trajectory and user-based collaborative filtering approach and implement a context-aware recommender system in order to provide the user with visiting routes in a museum. The system considers several contextual aspects, such as user preferences, choices of other visitors, time constraints, current location, and trajectory. To evaluate the proposed approach, they used a real dataset of the artwork of the Museum of Modern Art collection and reproduced the layout of six floors by converting map images available on the Web to graph structures by using the tool WebPlotDigitizer [64]. The DataGenCARS [65] tool was used for user rating generation, producing ratings provided by synthetic users for artworks already visited, based on the users' profiles, as well as random trajectories that were assigned to the users. The exchange of opinions among visitors relies on a central information service that allows feeding the recommendation process with data to pro-actively suggest changes in the recommended route in the museum. The recommendation approach detects visitors with similar preferences to the user and uses their ratings to estimate the potential ratings of the user for various artworks. If those ratings exceed a threshold, an artwork is considered a candidate for suggestion. The highest-rated items are ordered in a way that minimises the overall distance and the generated path that consists of their optimal sequence is suggested to the user.

DeepTrip is an end-to-end neural network method for understanding the underlying human mobility and modelling of the POI transitional distribution in human moving patterns. DeepTrip is proposed by Gao et al. [41] as an implementation of a trajectory embedding approach for a low dimensional representation of POI contextual features. A trip encoder that leverages a recurrent neural network is responsible for the route embeddings and a trip decoder for reconstructing the routes. A Generative Adversarial Network (GAN) is defined as a learning model consisting of a generator and a discriminator that compete in a two-player min-max game. The framework incorporates a GAN to enhance the generation ability of the trip decoder for POI sequence recommendations. Evaluation experiments show that DeepTrip outperforms the state-of-the-art baseline resulted from various evaluation metrics, although it expends more effort in understanding human mobility through learning implicit trajectory distributions.

Geo-tagged photos can be used to construct users' trajectories as they contain spatiotemporal information in sequential order and are useful for mining patterns of human movement. Cai et al. [42] present an itinerary RS based on semantic trajectory pattern mining from geo-tagged photos in order to provide a suggestion of POIs. Sequences of geotagged photos that capture POIs provide spatial and contextual information like weather conditions and travel duration. Semantic itineraries are built by annotating raw trajectories with application-dependent contextual and spatial semantics extracted from geo-tagged photos and environmental data like day, type, time, weather conditions, and place names. Then, the semantically enriched trajectories are mined by performing a variation of the Prefixspan algorithm for patterns of frequent sequences of semantic stops in the progress of the trajectories. The system consists of the offline trajectory pattern mining part and the online itinerary recommendation part. The recommendation part suggests itineraries based on a user query by filtering and ranking candidate itineraries from the semantic trajectory pattern database. The experimental results on real datasets from Flickr [66] support the effectiveness and efficiency of the proposed system over traditional approaches.

In Xu and Han [43], the authors introduce a framework for next location recommendation based on trajectory analysis and user behaviour. The framework is based on a Recurrent Neural Network (RNN) and a Similarity-based Markov Model (SMM) that combines inferred user behaviour and spatial information to suggest future locations. Raw trajectory data are converted offline to semantic sequences. Then, the users are added to clusters based on semantic similarity of their trajectories. The neural network is trained from the trajectory features of the user clusters. For the online location prediction part, the trajectory of the target user is transformed to a semantic sequence by applying the Word2vec algorithm [48] to embed spatiotemporal and semantic information into a universal space, and the user is attached to the most similar cluster. The SMM creates a state transition matrix from the correlation of the user with others in the cluster and historical trajectory data. The output of the model is an ordered sequence of candidate locations, where users can choose preferably. The proposed framework is reported to be superior to other tested models in terms of prediction performance.

5. Evaluation and Discussion

In Table 1, the evaluation of related works based on the requirements of the proposed framework is presented. The columns represent the requirements (as listed below) and the rows show the referenced studies. The order of the presented works in Table 1 is based on the structure and order of their presentation in Section 4. This rationale, along with the arrangement of the evaluation criteria (first half are related to ST, second half are related to RS), are the reasons for the effect in Table 1, i.e., quadrants I and III are comparatively empty.

*	a	b	c	d	e	f	g	h	i	j	k	1	m	n	0
[26]	x	x													
[13]	x	x	x	x											
[30]	x	x													
[31]	x	x													
[32]	x	x	x	x											
[33]	x	x													
[34]	х	x	x	x	x	x		x							
[35]	x	x				x									
[36]	x					x									
[37]	x	x	x		x	x									
[56]		x	x												
[27]						x	x								
[28]	x	x				x									
[29]	x	x		х		x	х								
[8]	x	x				x									
[44]								x						x	х
[45]				i	х			х	x	х		х		х	х
[46]			ii		x			x				х		х	х
[47]								х				х		х	х
[48]								x						х	
[49]								x							х
[50]								x	x			х	х	х	
[51]					x			x		x	x				
[38]								x		x	x				
[2]				iii	x			x	x	х		х	х	х	х
[39]								x	x	х	х	х			
[40]	x							x		х		х		х	x
[41]	x				x	x	х	x							x
[42]	x	x			x	x									x
[43]	х	х						х						х	

Table 1. Comparative table of reviewed related works.

i: enrichment of POIs with LOD, ii: ontological description of the context, iii: enrichment of POIs with LOD. * Based on the order presented in Section 4.

The main requirements of the framework are listed as follows:

- a. Exploitation of raw spatiotemporal trajectory data: raw trajectory data include useful spatial information combined with time-specific stops, speed, and direction of the visitor, needed for the initial segmentation.
- b. Semantic segmentation and annotation of the trajectory: for raw trajectories to be converted to semantic trajectories and analysed as such, segmentation of the trajectory and annotation of the parts are necessary.
- c. Trajectory description using suitable ontologies: Ontologies provided a structured and unified way of semantically describing instances of entities and fuse them with domain knowledge

- d. Semantic trajectory enrichment with linked open data (LOD): LOD grant a plethora of continuously updated information from different data sources regarding the context of the trajectory
- e. Semantic annotation of cultural spaces and points of interest (POI) to provide context and capability for semantic integration with user trajectories: semantically described POIs and spaces can make trajectory segmentation and recommendations more effective, and the interlinking of POIs and trajectories possible
- f. Trajectory analytics for pattern recognition and classification: the main goal of the process is the ability to discover features and recognise patterns in trajectories to categorise them and extract meaningful information about visitor movement
- g. Future location prediction: effective trajectory analysis and classification can lead to future location prediction, which is a useful input for the RSs
- h. Dynamic user profiling: updating user profile based on explicit and implicit feedback of user behaviour
- i. Integration of User Knowledge Graph (UKG): describes user profiles semantically and represents them as nodes in a KG
- j. Integration of Cultural Space (CS) and POI Knowledge Graph (KG): represents semantically annotated and enriched POIs and CS as a KG
- k. Integration of KG-Based recommender system (RS) for path-based and KG-based recommendations: performs path finding and connectivity methods for discovering possible recommendation lists in the optimal ranking order
- 1. Integration of context-aware RS: provides suggestions considering contextual information to enhance final recommendations
- m. Integration of hybrid RS: merges multiple recommendations to achieve maximum efficiency and accuracy
- n. Integration of collaborative filtering RS: leverages user similarity to produce meaningful suggestions
- o. Inference and proposal of a possible synthesis of visitor trajectories: evaluation and combination of RS suggestions with respect to user preferences for generating and proposing optimised trajectories

By reviewing the state-of-the-art method in semantic trajectories and recommender systems in recent literature, it appears that a significant amount of related work is partially related to the topics of semantic trajectories and semantic cultural recommender systems. However, as our research indicates, none of these related works fully exploit the semantic information and the insights of trajectory analytics to (a) enhance the functionality and efficiency of a recommender system for cultural spaces, and (b) connect user trajectories with cultural experiences. Some of the related works [40,41,46] propose an integration of RS approaches with trajectory extracted information, but either there is no semantic annotation or analysis of the trajectories, and the focus is only on GPS points, or the recommendation approach uses only the user current position and not the full semantic representation of user trajectory to define the context. Last but not least, in most of the related works [2,40–42,44–47,49], the outcome is a suggestion of a path generated based on user interest. That path usually is the shortest path of connected POIs, without considering the user visiting style.

6. Proposed System Architecture Design

In this paper, we present the architectural design of the proposed framework (Figure 4) along with a real-life use case scenario. The framework is designed to overcome the limitations of existing related work and fully meet the requirements to provide recommendations for optimum alternatives to visitor cultural experiences based on semantic trajectory analytics, as these were discussed in Section 5.



Figure 4. A system architecture design on the collaboration and information exchange between the Knowledge Graph and trajectory-oriented and recommendation-oriented components to provide optimal trajectory recommendations to the user.

The framework is designed to be trajectory-centred and to express the integration of user experience and movement in an enriched semantic trajectory. Results of analytics on that trajectory and the use of information of similar trajectories in the same cluster will provide valuable input for meaningful recommendations. Furthermore, the suggestions for the future visiting locations and the information about POIs are designed to be tailored to user preferences and continuously re-evaluated based on user choices and their physical location. As a trajectory evolves, it can be classified in a more effective manner and can be more accurately matched to the visiting style(s) of a user.

6.1. Use Case Scenario

The recording of the purpose of a cultural trip to a city is a semantic annotation at the trajectory level, while the recording of the presence of a person at a specific location, such as a visit to a temporal art exhibition, is a semantic annotation at the position level. Movement analysis of a tourist who engages in cultural related activities in Athens results in a trajectory for the whole movement in the city as a 'tourist inside Athens'. One or more distinct trajectories of daily cultural experiences, such as a tour of the Museum of Acropolis on Friday morning, are also recorded. The distinct areas of the museum and the exhibits are semantically described. Visitor data, such as age and gender, personal preferences regarding art, music, types of museums a user visits, and the ways of touring in an exhibition space have been collected and evaluated. During the visit to the Acropolis Museum, that information is combined to produce semantically richer and more accurate trajectories, as well as to suggest routes and alternatives to improve the visitor experience. Trajectory-based personal recommendations are provided via a smartphone application on a user's phone screen, or in embedded screens near exhibits or artworks. The recommendations provide either short or detailed descriptions based on calculated user profiles and suggestions for relevant and related exhibits along with the most efficient routes to reach them.

The tourist in our scenario uses the application on a smartphone device. The application records the GPS signal to monitor the current location of the user (Trajectory Monitoring). The user has provided personal information and interests to build an initial version of a profile. As depicted in Figure 1, the user starts the walk from Plaka and, after a while, stops for a coffee break. This information is stored and used to annotate the trajectory with preferences in sightseeing and social activities (Figure 4: Trajectory Segmentation, Trajectory Annotation). As the user moves in Athens, the trajectory is taking shape and it is compared with stored trajectories in order to be grouped in a cluster with trajectories that have similar characteristics (Figure 4: Trajectory Clustering). The analysed trajectory is classified as a 'touristic walk inside Athens' (Figure 4: Trajectory Classification). The user gets a notification from the application to continue the walk to Acropolis to visit the Parthenon and receives information about this POI. The user provides feedback for the suggestion and evaluates the related information (Figure 4: explicit User Feedback). The system discovers that similar trajectories follow the path to Acropolis Museum, so it suggests the museum as the next visiting location (Figure 4: Collaborative Filtering Recommendations). The user is provided with a basic path that covers the main exhibits of the museum, based on their current profile preferences. Inside the museum, Bluetooth beacons, installed near the exhibits, provide information to the application about the proximity of the user and the time spent near them. That information, along with the interaction of the user to the provided contextual details about the targeted exhibit, are recorded as interesting information for them (Figure 4: Implicit User Feedback, Dynamic Profiling). The system is actively computing new recommendations and provides them to the user. The recommendations are based on the user preferences, the available time span, and the visiting style. The visiting style is evaluated by comparing the current trajectory with others stored in the KG. Data from the beacons and the application show that the user expresses more interest in statues rather than other exhibits. The application suggests that visitors with this preference follow a route from the Archaic Acropolis on the 1st floor to the Caryatids and the Athena Nike on the 2nd floor (Figure 4: Collaborative Filtering Recommendations). The recommendations consider avoiding crowded exhibits and rearrange the order of the suggestions (Figure 4: Context-Aware Recommendations). The calculated tour in the museum is estimated to last approximately two hours. After the first hour, the user is detected to behave differently, i.e., not devoting enough attention to the exhibits, bypassing most of the system suggestions. This behaviour is recognised as boredom or exhaustion (Figure 4: Behaviour Monitoring, Experience Evaluation). The application re-evaluates the suggested route and proposes the following (in sequence): (a) to skip the following exhibits on the 2nd floor, (b) to move to the 3rd floor, (c) to take a 20-min break at the museum café, and (d) to visit the temporal exhibition of the Acropolis Museum. The suggestion about the 3rd floor, where the artworks from the Parthenon are exhibited, was based on the interest and positive feedback received from the user about the Parthenon earlier in the tour (Figure 4: Knowledge-based/Context-Aware Recommendations).

6.2. System Architecture Modules and Tasks

In Figure 4, the proposed system architecture design for the proposed framework is depicted. The framework is divided to interconnected modules and information exchangers. The distinct modules are Monitoring, Pre-processing, Semantic Representation, Profiling, Semantic Trajectories, Trajectory Analytics, and Recommendation.

The Monitoring Module includes trajectory and user behaviour monitoring tasks. It is assumed that the cultural space is equipped with proper monitoring infrastructure (e.g., IoT devices such as Bluetooth beacons or RFIDs, and cameras) for tracking the visitor movement and behaviour. The Trajectory Monitoring task is responsible for collecting information about the physical movement of the user. This information is generated from GPS signals of portable devices such as smartphones and smartwatches, and proximity signals from Bluetooth beacons or RFID tags. The User Behaviour Monitoring task collects information about the behaviour of the user. This information is collected from the interactions of users with POIs, recorded by the developed application or by the evaluation of the user's mood (e.g., bored, interested). Movement and facial expressions can also be recorded by cameras in inner (e.g., museum) or open spaces (e.g., open museums/archaeological sites).

The Pre-processing Module receives data from the Trajectory Monitoring and Behaviour Monitoring tasks. At this module, the data is cleaned and integrated for further processing. Data integration is implemented by semantically describing the data with suitable ontologies. For instance, a data model for the system could use the datAcron Ontology [54] to describe trajectories, EDM [67] or CIDOC-CRM [68] to describe artworks, and FOAF [69] or User Profile Ontology [70] to describe users. Data is cleaned and transformed to RDF triples with tools like Karma [71] and eventually stored as a KG. This data conversion is necessary to (a) handle the heterogeneous data in a unified manner, and (b) for further enrichment and linking with related LOD.

The Semantic Trajectories Module is responsible for the transformation of raw trajectories to semantic trajectories. After data transformation, the trajectories formed by raw spatiotemporal data are segmented based on stop/move parts, velocity, or predefined POIs. The segments are semantically annotated with contextual information (day, time, place name, weather, etc.) and domain knowledge (e.g., abstract concept about the artwork). The semantic trajectories are also enriched with LOD (e.g., cultural POIs linked with a DBpedia or Europeana entity), complementary information stored in the KG, such as data provided by the Semantic Representation Module, or features visiting style evaluated by the comparison of stored trajectories in the KG.

The Semantic Representation Module is responsible for the representation of the cultural space. To provide predefined POIs and ROIs for more efficient trajectory segmentation, the cultural space and its exhibits must be described in a manner that is in alignment with the movement data and persist useful features of artworks and their peripheral area. POIs and ROIs are then semantically described with a suitable ontology that covers semantic and spatial information, converted to RDF, and stored as main entities to the cultural space KG. The entities are now linked and enriched with external information from LOD. Furthermore, they are interlinked with the POIs recognised in the trajectory segmentation part.

The Trajectory Analytics Module is responsible for the analysis and categorization of semantic trajectories. It consists of two tasks: the "online" Real-Time Trajectory Classification and the "offline" Trajectory Clustering task. The stored semantic trajectories are then analysed for pattern extraction and for the formation of semantically similar trajectory clusters. Each cluster contains trajectories with discovered visiting styles that are spatially or semantically similar. For example, one may spend the whole visit exploring all artworks and artifacts of a temporary exhibition while another is selectively visiting the essential and mainstream exhibits of a museum and spend the rest of the visit in the museum cafe. Clusters are necessary for the "online" part, where the partially constructed trajectories are assigned to the most similar one in real-time and provide to the semantic segmentation and annotation part useful information about entities' visiting style and preferences. The Profiling Module is responsible for updating the user profile and evaluating user state. It consists of the Feedback task and the Dynamic Profiling and Experience Evaluation task. Implicit (user actions in the monitored space, visited exhibits and time spent near them) and explicit (direct ratings through forms/questionnaires) user feedback, alongside the classified trajectory, provide essential information for dynamic profiling. The Dynamic Profiling & Experience Evaluation task combines information about user preferences with feedback and behaviour monitoring data to evaluate user experience and dynamically update the user profile. The recording profiles, containing personal information and provided or inferred preferences, are stored in the User KG.

The Recommendation Module consists of the Collaborative Filtering task, the Knowledgebased/Context-Aware recommendations task, and the Trajectory synthesis and Recommendation of optimal trajectories task. CF RS methods applied to the User KG provide suggestion lists based on user similarity. A Knowledge-based/Context-aware RS provides suggestions based on semantic object similarity measurements and user preferences while considering contextual information like crowd density, weather, and minimum time needed to explore a region. This module is also responsible for providing complementary semantic and multimedia information about the visited artifacts. For instance, the user (e.g., Peter in Figure 3) is interested in a specific artwork that is described in the KG (Caryatid). The artwork is linked with information and related entities in the KG, such as the creator and the museum that it is exhibited at. The museum (Acropolis Museum) is related to the city (Athens) that it is located, while the city is related to its POIs. If other users with similar preferences (Mary) have shown interest in POIs (Pantheon) in this city, a recommended path will occur from one POI to the other, based on the feature similarity, community detection, and reachability derived from the KG.

The Trajectory synthesis and Recommendation of optimal trajectories task integrates a hybrid RS that merges the recommendation provided by the CF RS and the Knowledgebased RS. This RS leverages the merged recommendations to achieve trajectory synthesis and provide optimal personalised routes that guide visitors to preferred exhibits, according to the preferred visiting style and the available timespan. A crucial part of the described architecture is the continuous feedback and the experience evaluation that affect the RS state and create the potential to dynamically update the provided recommendations and the complementary information.

The proposed architecture is designed to meet the abovementioned requirements and cover the partial absence of ST exploitation for empowering RS towards cultural experience optimisation. Furthermore, the architecture is designed to capture the entire process of handling raw spatiotemporal data, converting them to enriched ST, clustering and classifying ST based on features, creating user and cultural space KG, and integrating different types of RS to effectively recommend enhanced trajectories.

7. Conclusions and Future Work

Cultural spaces like museums are increasingly emphasising a more personalised, optimised, and enhanced visiting experience. A way to achieve that is through efficiently and effectively understanding human movement in cultural spaces. Movement can be effectively represented and evaluated by semantic trajectory analysis, while personalisation can be realised by user/visitor profiling and by providing meaningful and interesting suggestions utilising specialised, software-like recommender systems. In this paper, we conducted a systematic review of the state-of-the-art related work, focusing on the intersection of the semantic trajectories and recommender systems, and on the advantages of the Semantic Web Technologies and KGs in both research fields. Subsequently, a framework and a system for the collection, annotation, and analysis of trajectory data, along with the integration of a hybrid knowledge-based RS, is proposed for optimising cultural experiences. Future plans for this work involve the implementation of the proposed system by developing a set of methods and tools that meet the presented framework requirements by (a) the effective transformation of raw trajectories to semantic trajectories, (b) the performing of analytic tasks to extract meaningful information about visitors, and (c) the integration of a hybrid RS that combines results from a KG-based, a CF-based, and context-aware RS, for optimal suggestions. Future work will also include the design and evaluation of a use case scenario and experimentation with real-life users to receive useful feedback on the efficiency of the framework.

Author Contributions: Conceptualisation: S.A. and K.K.; methodology: K.K.; formal analysis: S.A.; investigation: S.A.; resources: D.S.; writing—original draft preparation: S.A.; writing—review and editing: K.K. and D.S.; visualisation: S.A.; supervision: K.K. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors would like to thank Alexandra Kallitsa for her valuable contribution.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Ruotsalo, T.; Haav, K.; Stoyanov, A.; Roche, S.; Fani, E.; Deliai, R.; Mäkelä, E.; Kauppinen, T.; Hyvönen, E. SMARTMUSEUM: A mobile recommender system for the Web of Data. *J. Web Semant.* **2013**, *20*, 50–67. [CrossRef]
- 2. Sansonetti, G.; Gasparetti, F.; Micarelli, A.; Cena, F.; Gena, C. Enhancing cultural recommendations through social and linked open data. *User Model. User-Adapt. Interact.* **2019**, *29*, 121–159. [CrossRef]
- Van Hage, W.R.; Stash, N.; Wang, Y.; Aroyo, L. Finding your way through the Rijksmuseum with an adaptive mobile museum guide. In Proceedings of the 7th Extended Semantic Web Conference, ESWC 2010, Heraklion, Greece, 30 May–3 June 2010; Volume 9088, pp. 46–59. [CrossRef]
- Andrienko, G.; Andrienko, N.; Fuchs, G.; Raimond, A.M.O.; Symanzik, J.; Ziemlicki, C. Extracting semantics of individual places from movement data by analyzing temporal patterns of visits. In Proceedings of the First ACM SIGSPATIAL International Workshop on Computational Models of Place, Orlando, FL, USA, 5–8 November 2013; pp. 9–15. [CrossRef]
- 5. Zhang, D.; Lee, K.; Lee, I. Hierarchical trajectory clustering for spatio-temporal periodic pattern mining. *Expert Syst. Appl.* **2018**, 92, 1–11. [CrossRef]
- Ying, J.J.C.; Lu, E.H.C.; Lee, W.C.; Weng, T.C.; Tseng, V.S. Mining user similarity from semantic trajectories. In Proceedings of the 2nd ACM SIGSPATIAL International Workshop on Location Based Social Networks (LBSN-10), San Jose, CA, USA, 2 November 2010; pp. 19–26. [CrossRef]
- Giannotti, F.; Nanni, M.; Pedreschi, D.; Pinelli, F.; Renso, C.; Rinzivillo, S.; Trasarti, R. Unveiling the complexity of human mobility by querying and mining massive trajectory data. VLDB J. 2011, 20, 695–719. [CrossRef]
- 8. Liu, S.; Wang, S. Trajectory Community Discovery and Recommendation by Multi-Source Diffusion Modeling. *IEEE Trans. Knowl. Data Eng.* **2017**, *29*, 898–911. [CrossRef]
- 9. Parent, C.; Spaccapietra, S.; Renso, C.; Andrienko, G.; Andrienko, N.; Bogorny, V.; Damiani, M.L.; Gkoulalas-Divanis, A.; Macedo, J.; Pelekis, N.; et al. Semantic trajectories modeling and analysis. *ACM Comput. Surv.* **2013**, *45*, 1–32. [CrossRef]
- 10. Spaccapietra, S.; Parent, C.; Damiani, M.L.; de Macedo, J.A.; Porto, F.; Vangenot, C. A conceptual view on trajectories. *Data Knowl. Eng.* 2008, 65, 126–146. [CrossRef]
- Nanni, M.; Trasarti, R.; Renso, C.; Giannotti, F.; Pedreschi, D. Advanced knowledge discovery on movement data with the GeoPKDD system. In Proceedings of the 13th International Conference on Extending Database Technology, Lausanne, Switzerland, 22–26 March 2010; pp. 693–696. [CrossRef]
- 12. Bao, J.; Zheng, Y.; Wilkie, D.; Mokbel, M. Recommendations in location-based social networks: A survey. *Geoinformatica* 2015, *19*, 525–565. [CrossRef]
- 13. Nogueira, T.P.; Braga, R.B.; de Oliveira, C.T.; Martin, H. FrameSTEP: A framework for annotating semantic trajectories based on episodes. *Expert Syst. Appl.* **2018**, *92*, 533–545. [CrossRef]
- Maarala, A.I.; Su, X.; Riekki, J. Semantic Reasoning for Context-Aware Internet of Things Applications. *IEEE Internet Things J.* 2017, 4, 461–473. [CrossRef]
- 15. Dodge, S.; Weibel, R.; Lautenschütz, A.K. Towards a taxonomy of movement patterns. Inf. Vis. 2008, 7, 240–252. [CrossRef]
- 16. Kembellec, G.; Chartron, G.; Saleh, I. Recommender Systems; John Wiley & Sons: Hoboken, NJ, USA, 2014; ISBN 9781119054252.

- 17. Pavlidis, G. Recommender systems, cultural heritage applications, and the way forward. J. Cult. Herit. 2019, 35, 183–196. [CrossRef]
- 18. Bobadilla, J.; Ortega, F.; Hernando, A.; Gutiérrez, A. Recommender systems survey. Knowl.-Based Syst. 2013, 46, 109–132. [CrossRef]
- 19. Ricci, F.; Rokach, L.; Shapira, B. Recommender Systems Handbook; Springer: Berlin/Heidelberg, Germany, 2011; ISBN 9780387858203.
- 20. Barranco, M.J.; Noguera, J.M.; Castro, J.; Martínez, L. A context-aware mobile recommender system based on location and trajectory. *Adv. Intell. Syst. Comput.* **2012**, *171 AISC*, 153–162. [CrossRef]
- 21. Chicaiza, J.; Valdiviezo-Diaz, P. A comprehensive survey of knowledge graph-based recommender systems: Technologies, development, and contributions. *Information* **2021**, *12*, 232. [CrossRef]
- 22. Hogan, A.; Blomqvist, E.; Cochez, M.; D'Amato, C.; De Melo, G.; Gutierrez, C.; Kirrane, S.; Gayo, J.E.L.; Navigli, R.; Neumaier, S.; et al. Knowledge graphs. *ACM Comput. Surv.* 2021, 54, 1–257. [CrossRef]
- 23. Bonatti, P.; Decker, S.; Polleres, A.; Presutti, V. Knowledge Graphs: New Directions for Knowledge Representation on the Semantic Web (Dagstuhl Seminar 18371). *Dagstuhl Rep.* **2019**, *8*, 29–111.
- 24. Kejriwal, M. What Is a Knowledge Graph. In *Domain-Specific Knowledge Graph Construction;* SpringerBriefs in Computer Science; Springer: Cham, Switzerland, 2019. [CrossRef]
- Lassila, O.; Swick, R.R. Resource Description Framework (RDF) Model and Syntax Specification. World Wide Web Consortium Recommendation. 1999. Available online: https://www.w3.org/TR/1999/REC-rdf-syntax-19990222/ (accessed on 16 November 2021).
- De Graaff, V.; De By, R.A.; De Keulen, M. Automated semantic trajectory annotation with indoor point-of-interest visits in urban areas. In Proceedings of the 31st Annual ACM Symposium on Applied Computing, Pisa, Italy, 4–8 April 2016; pp. 552–559. [CrossRef]
- 27. Chen, Z.; Wang, X.; Li, H.; Wang, H. On Semantic Organization and Fusion of Trajectory Data. In Proceedings of the 2020 IEEE 44th Annual Computers, Software, and Applications Conference (COMPSAC), Madrid, Spain, 13–17 July 2020; pp. 1078–1081. [CrossRef]
- 28. Al-Dohuki, S.; Wu, Y.; Kamw, F.; Yang, J.; Li, X.; Zhao, Y.; Ye, X.; Chen, W.; Ma, C.; Wang, F. SemanticTraj: A New Approach to Interacting with Massive Taxi Trajectories. *IEEE Trans. Vis. Comput. Graph.* **2017**, *23*, 11–20. [CrossRef]
- Santipantakis, G.M.; Glenis, A.; Patroumpas, K.; Vlachou, A.; Doulkeridis, C.; Vouros, G.A.; Pelekis, N.; Theodoridis, Y. SPARTAN: Semantic integration of big spatio-temporal data from streaming and archival sources. *Futur. Gener. Comput. Syst.* 2020, 110, 540–555. [CrossRef]
- Soares, A.; Times, V.; Renso, C.; Matwin, S.; Cabral, L.A.F. A semi-supervised approach for the semantic segmentation of trajectories. In Proceedings of the 2018 19th IEEE International Conference on Mobile Data Management (MDM), Aalborg, Denmark, 25–28 June 2018; pp. 145–154. [CrossRef]
- 31. Vassilakis, C.; Kotis, K.; Spiliotopoulos, D.; Margaris, D.; Kasapakis, V.; Anagnostopoulos, C.N.; Santipantakis, G.; Vouros, G.A.; Kotsilieris, T.; Petukhova, V.; et al. A semantic mixed reality framework for shared cultural experiences ecosystems. *Big Data Cogn. Comput.* **2020**, *4*, 6. [CrossRef]
- 32. Ghosh, S.; Ghosh, S.K. Modeling of human movement behavioral knowledge from GPS traces for categorizing mobile users. In Proceedings of the 26th International Conference on World Wide Web Companion, Perth, Australia, 3–7 April 2017; pp. 51–58. [CrossRef]
- 33. Gao, C.; Zhang, Z.; Huang, C.; Yin, H.; Yang, Q.; Shao, J. Semantic trajectory representation and retrieval via hierarchical embedding. *Inf. Sci.* (*NY*) **2020**, *538*, 176–192. [CrossRef]
- Kontarinis, A.; Zeitouni, K.; Marinica, C.; Vodislav, D.; Kotzinos, D. Towards a semantic indoor trajectory model: Application to museum visits. *GeoInformatica* 2021, 25, 311–352. [CrossRef] [PubMed]
- Karatzoglou, A.; Schnell, N.; Beigl, M. A convolutional neural network approach for modeling semantic trajectories and predicting future locations. In Proceedings of the 27th International Conference on Artificial Neural Networks, Rhodes, Greece, 4–7 October 2018; Volume 11139, pp. 61–72. [CrossRef]
- 36. Zhang, W.; Wang, X.; Huang, Z. A system of mining semantic trajectory patterns from GPS data of real users. *Symmetry* **2019**, *11*, 889. [CrossRef]
- 37. Khoroshevsky, F.; Lerner, B. Human mobility-pattern discovery and next-place prediction from GPS data. In Proceedings of the 4th IAPR TC 9 Workshop, MPRSS 2016, Cancun, Mexico, 4 December 2016; Volume 10183, pp. 24–35. [CrossRef]
- 38. Amato, F.; Moscato, F.; Moscato, V.; Pascale, F.; Picariello, A. An agent-based approach for recommending cultural tours. *Pattern Recognit. Lett.* **2020**, *131*, 341–347. [CrossRef]
- Su, X.; Sperl, G.; Moscato, V.; Picariello, A. An Edge Intelligence Empowered Recommender System Enabling Cultural Heritage Applications. *IEEE Trans. Ind. Inform.* 2019, 15, 4266–4275. [CrossRef]
- 40. Cardoso, P.J.S.; Rodrigues, J.M.F.; Pereira, J.; Nogin, S.; Lessa, J.; Ramos, C.M.Q.; Bajireanu, R.; Gomes, M.; Bica, P. Cultural heritage visits supported on visitors' preferences and mobile devices. *Univers. Access Inf. Soc.* **2020**, *19*, 499–513. [CrossRef]
- 41. Smirnov, A.V.; Kashevnik, A.M.; Ponomarev, A. Context-based infomobility system for cultural heritage recommendation: Tourist Assistant—TAIS. *Pers. Ubiquitous Comput.* **2017**, *21*, 297–311. [CrossRef]
- 42. Hong, M.; An, S.; Akerkar, R.; Camacho, D.; Jung, J.J. Cross-cultural contextualisation for recommender systems. J. Ambient Intell. Humaniz. Comput. 2019, 10, 1–12. [CrossRef]
- Loboda, O.; Nyhan, J.; Mahony, S.; Romano, D.M.; Terras, M. Content-based Recommender Systems for Heritage: Developing a Personalised Museum Tour. In Proceedings of the DSRS-Turing 2019: 1st International 'Alan Turing' Conference on Decision Support and Recommender Systems, London, UK, 21–22 November 2019.
- 44. Hong, M.; Jung, J.J.; Piccialli, F.; Chianese, A. Social recommendation service for cultural heritage. *Pers. Ubiquitous Comput.* **2017**, 21, 191–201. [CrossRef]

- 45. Qassimi, S.; Abdelwahed, E.H. Towards a semantic graph-based recommender system. A case study of cultural heritage. *J. Univers. Comput. Sci.* **2021**, *27*, 714–733. [CrossRef]
- Zhou, S.; Dai, X.; Chen, H.; Zhang, W.; Ren, K.; Tang, R.; He, X.; Yu, Y. Interactive Recommender System via Knowledge Graph-enhanced Reinforcement Learning. In Proceedings of the 43rd International ACM SIGIR Conference on Research and Development in Information Retrieval, Virtual Event, China, 25–30 July 2020; pp. 179–188. [CrossRef]
- 47. Minkov, E.; Kahanov, K.; Kuflik, T. Graph-based recommendation integrating rating history and domain knowledge: Application to on-site guidance of museum visitors. *J. Assoc. Inf. Sci. Technol.* **2017**, *68*, 1911–1924. [CrossRef]
- 48. Rodríguez-Hernández, M.D.C.; Ilarri, S.; Hermoso, R.; Trillo-Lado, R. Towards Trajectory-Based Recommendations in Museums: Evaluation of Strategies Using Mixed Synthetic and Real Data. *Procedia Comput. Sci.* **2017**, *113*, 234–239. [CrossRef]
- 49. Gao, Q.; Zhou, F.; Zhang, K.; Zhang, F.; Trajcevski, G. Adversarial Human Trajectory Learning for Trip Recommendation. *IEEE Trans. Neural Netw. Learn. Syst.* 2021, 32, 1–13. [CrossRef]
- 50. Cai, G.; Lee, K.; Lee, I. Itinerary recommender system with semantic trajectory pattern mining from geo-tagged photos. *Expert Syst. Appl.* **2018**, *94*, 32–40. [CrossRef]
- 51. Xu, M.; Han, J. Next Location Recommendation Based on Semantic-Behavior Prediction. In Proceedings of the 2020 5th International Conference on Big Data and Computing, Chengdu, China, 28–30 May 2020; pp. 65–73. [CrossRef]
- 52. Semantic Trajectory Episodes—Report Generated by Parrot. Available online: http://talespaiva.github.io/step/ (accessed on 16 November 2021).
- 53. OpenStreetMap. Available online: https://www.openstreetmap.org/#map=16/37.9704/23.7300&layers=H (accessed on 16 November 2021).
- 54. Santipantakis, G.M.; Vouros, G.A.; Doulkeridis, C.; Vlachou, A.; Andrienko, G.; Andrienko, N.; Fuchs, G.; Garcia, J.M.C.; Martinez, M.G. Specification of semantic trajectories supporting data transformations for analytics: The datacron ontology. In Proceedings of the 13th International Conference on Semantic Systems, Amsterdam, The Netherlands, 11–14 September 2017; pp. 17–24. [CrossRef]
- 55. IndoorGML OGC. Available online: http://indoorgml.net/ (accessed on 16 November 2021).
- 56. Krisnadhi, A.; Hitzler, P.; Janowicz, K. A spatiotemporal extent pattern based on semantic trajectories. *Adv. Ontol. Des. Patterns* **2017**, *32*, 47–53.
- 57. Pei, J.; Han, J.; Mortazavi-Asl, B.; Pinto, H.; Chen, Q.; Dayal, U.; Hsu, M.C. PrefixSpan: Mining sequential patterns efficiently by prefix-projected pattern growth. In Proceedings of the 17th International Conference on Data Engineering, Heidelberg, Germany, 2–6 April 2001; pp. 215–224. [CrossRef]
- 58. Graph Data Platform | Graph Database Management System | Neo4j. Available online: https://neo4j.com/ (accessed on 16 November 2021).
- 59. Home—DBpedia Association. Available online: https://www.dbpedia.org/ (accessed on 16 November 2021).
- 60. Discover Inspiring European Cultural Heritage | Europeana. Available online: https://www.europeana.eu/en (accessed on 16 November 2021).
- 61. Home—LinkedGeoData. Available online: http://linkedgeodata.org/ (accessed on 16 November 2021).
- 62. SPARQL 1.1 Query Language. Available online: https://www.w3.org/TR/sparql11-query/ (accessed on 16 November 2021).
- 63. Haveliwala, T.H. Topic-sensitive PageRank. In Proceedings of the Eleventh International Conference on World Wide Web—WWW '02, Honolulu, HI, USA, 7–11 May 2002; ACM Press: New York, NY, USA, 2002; p. 517.
- 64. WebPlotDigitizer—Extract Data from Plots, Images, and Maps. Available online: https://automeris.io/WebPlotDigitizer/ (accessed on 16 November 2021).
- 65. DataGenCARS. Available online: http://webdiis.unizar.es/~{}silarri/DataGenCARS/ (accessed on 16 November 2021).
- 66. Find Your Inspiration. | Flickr. Available online: https://flickr.com/ (accessed on 16 November 2021).
- 67. Europeana Data Model | Europeana Pro. Available online: https://pro.europeana.eu/page/edm-documentation (accessed on 16 November 2021).
- 68. Home | CIDOC CRM. Available online: http://www.cidoc-crm.org/ (accessed on 16 November 2021).
- 69. FOAF Vocabulary Specification. Available online: http://xmlns.com/foaf/spec/ (accessed on 16 November 2021).
- 70. User Profile Ontology. Available online: http://iot.ee.surrey.ac.uk/citypulse/ontologies/up/up.html (accessed on 16 November 2021).
- 71. Karma: A Data Integration Tool. Available online: https://usc-isi-i2.github.io/karma/ (accessed on 16 November 2021).